Author's response to reviews

Title: Over-diagnosed and over-treated? A survey of Australian public attitudes towards the acceptability of drug treatment for depression and ADHD.

Authors:

Bradley Partridge (b.partridge@uq.edu.au)
Jayne Lucke (j.lucke@uq.edu.au)
Wayne Hall (w.hall@uq.edu.au)

Version: 3 Date: 26 February 2014

Author's response to reviews:

Dear Editor,

The following changes have now been made in the manuscript as requested.

1. A 'Conclusions' section is now included on page 16.

2. “Declaration of Conflicting Interests” has been changed to “Competing interests” on page 17 and the funding acknowledgement is now included in the “Acknowledgements” section.

3. “Authors’ contribution” states:
“BP, JL, and WH contributed to the design of the study and creation of the survey items. BP analysed the data. BP, JL, and WH contributed to the drafting of the manuscript. All authors read and approved the final manuscript.”

4. The full name of the ethics committee that approved the study is now included in the methods section: “Ethics approval for the study was granted by the CQU Human Research Ethics Committee.”

Kind regards.